Author's response to reviews

Title: Alcohol use among university students in Sweden measured by an electronic screening instrument

Authors:

Agneta Andersson (agneta.andersson@liu.se)
Ann-Britt Wiréhn (ann-britt.wirehn@lio.se)
Christina Ölvander (christina.olvander@logica.com)
Diana Stark Ekman (dianastarkekman@gmail.com)
Preben Bendtsen (preben.bendtsen@liu.se)

Version: 5 Date: 24 June 2009

Author's response to reviews:

Dear Editors,

I am asked by first author Agneta Andersson to make the changes you have requested. I have attempted to upload a new manuscript, incorporating your suggestions.

1. You ask: Author list - The list of authors in the manuscript should be written exactly as they are in the submission system. The authors (forename, initial (if any), surname) should be written identically and in the order in which they are listed in the submission system. Currently the first author is listed as 'Andersson A' in the manuscript and 'Agneta Andersson' in the submission system.

We say: We have changed the author formatting on the paper to include full first name followed by last name, to match the online system.

2. You say: Table citation - All tables must be cited in the manuscript in consecutive order. Currently, tables 1-4 are not cited, whereas there is a table 5 and a table 6 cited in the manuscript when only 4 tables have been uploaded.

We say: Thank you-without your careful checking, we would not have known that two tables were missing in our manuscript!! We are grateful to the sharp editorial staff. We have adjusted the manuscript so that there is text citing all SIX tables.

I hope that the changes meet your satisfaction.

On behalf of all authors,

Diana Stark Ekman